Author's response to reviews

Title: Right atrium myxoma coexisting with antiphospholipid syndrome: A case report.

Authors:

Christos Pliakos (pliakos@med.auth.gr)
Eleni Alexiadou (alexiadou@yahoo.gr)
Symeon Metallidis (metallidissimeon@yahoo.gr)
Theodosis Papavramidis (papavramidis@hotmail.com)
Stergios Kapoulas (kapoulas@med.auth.gr)
Konstantinos Sapalidis (sapalidisk@yahoo.gr)
Pavlos Nikolaidis (nikolp@med.auth.gr)

Version: 2 Date: 21 September 2009

Author's response to reviews:

Dear Sir

Changes have been made according to the reviewers suggestions.

All changes are highlighted in the manuscript

A paragraph has been added in the discussion giving a possible explanation regarding the onset symptoms of the patient and the connection between myxoma and antiphospholipid syndrome.

Also an MRI of the myxoma is provided